Author's response to reviews

Title: War-related psychological sequelae among emergency department patients in the former Republic of Yugoslavia

Authors:

Brett D Nelson BA (bnelso1@jhmi.edu)
Dr William G Fernandez MD, MPH (william.fernandez@bmc.org)
Dr Sandro Galea MD, DrPH (sgalea@nyam.org)
Sarah Sisco MPH, MSSW (ssisco@nyam.org)
Kerry Dierberg BS (kzierbe1@jhmi.edu)
Gordana SUBARIC Gorgieva MD, PhD (subariogorgieva@yahoo.com)
Arijit K Nandi BS (akn2005@columbia.edu)
Jennifer Ahern MPH (jahern@nyam.org)
Dr Mihajlo Mitrovic MD (gvozden@eunet.yu)
Michael VanRooyen MD, MPH (mvanrooy@jhmi.edu)
David Vlahov PhD (dvlahov@nyam.org)

Version: 5 Date: 5 May 2004

PDF covering letter
Re: MS: 7642205492459296

Brett D Nelson, William G Fernandez, Sandro Galea, Sarah Sisco, Kerry Dierberg, Gordana Subaric Gorgieva, Arijit K Nandi, Jennifer Ahern, Mihajlo Mitrovic, Michael VanRooyen, David Vlahov: War-related psychological sequelae among emergency department patients in the former Republic of Yugoslavia, resubmitted to BMC Medicine

Dear Dr Gadd,

We are submitting the formatting revisions as requested for publication in BMC Medicine. Below are the point-by-point changes made:

Title page:

Qualifications (PhD, MD, etc, have been deleted)

We have removed all unnecessary capitalization of Section Headings.

Abstract Section and Main body of text:

We have removed all unnecessary capitalization of Section Headings.

We have changed the Heading “Conclusion” to “Conclusions”

We have made the appropriate formatting changes to the references as requested.

Tables:

We have made the appropriate formatting changes to the tables as requested.

--------

Thank you very much for the opportunity to work with you on this paper.

Regards,

--Bill Fernandez, MD, MPH